Author's response to reviews

Title: Giant scrotal elephantiasis of inflammatory etiology: Case report

Authors:

Stefan Denzinger (stefandenzinger@gmx.de)
Elke Watzlawek (watzlawek@caritasstjosef.de)
Maximilian Burger (maximilianburger@gmx.de)
Wolf F Wieland (wieland@caritasstjosef.de)
Wolfgang Otto (wolfgang1.otto@klinik.uni-regensburg.de)

Version: 3 Date: 12 May 2007

Author’s response to reviews:

Journal of Medical Case Reports
The Journal of Medical Case Reports
Editorial Team

Dear Editorial Team,

you asked to readdress the standard of English and to check format and references of our manuscript in accordance to the J Med Case Rep style. Please find enclosed the revised version of our case report: "Giant scrotal elephantiasis of inflammatory etiology: Case Report". We have proved and corrected format and references of our manuscript and let check the text by a native speaker.

We hope that the revised manuscript version is suitable for publication in Journal of Medical Case Reports.

Yours sincerely,

Wolfgang Otto, for all authors